

Author's response to reviews

Title: Scoliosis treatment using a combination of manipulative and rehabilitative therapy: a retrospective case series

Authors:

Mark W Morningstar (drmark2star@yahoo.com)

Dennis Woggon (dwoggon@aol.com)

Gary Lawrence (healthpros@juno.com)

Version: 3 **Date:** 4 September 2004

Author's response to reviews: see over

To Whom It May Concern:

As advised in the recent editorial comments, several format issues with the manuscript have been revised and corrected.

1. The title has been corrected
2. The headings in the paper have been corrected
3. All competing interests for all 3 authors have been fully declared
4. The authors' affiliations have been revised
5. The authors' qualifications have been removed
6. The references have been updated in the body of the manuscript
7. All figures have been cropped and formatted per editorial instruction
8. Figure legends were added to end of paper
9. The table was added to the end of the paper in portrait format
10. Additional table file will be removed
11. Per reviewer comments, lines were added to first image set in Figure 5

Reference 12 is still waiting to be published. It is scheduled to be published in Dec 2004.

Finally, the signed consent letters will be faxed to the BioMed Central editorial office when I get back to my clinic on Wed, 9/8/04. I will fax them to Matt Hodgkinson at +44 (0) 20 7636 4296

Regards,

Mark Morningstar